Author’s response to reviews

Title: Lower Vitamin D Levels in Saudi Pregnant Women are Associated with Higher Risk of Developing GDM

Authors:

Abdulrahman Al-AJlan (aalajl@hotmail.com)
Sara Almusharaf (sara.almosharruf@gmail.com)
Mona Fouda (monafoudaneel@yahoo.com)
Soundararajan Krishnaswamy (ksounder@gmail.com)
Naji Aljohani (najij@hotmail.com)
Amal Al-Serehi (aalserehi@kfmc.med.sa)
Eman Sheshah (eman_shesha@hotmail.com)
Naemah Alshingetti (alhingetti@yahoo.com)
Iqbah Turkistani (iqbalzmt@hotmail.com)
Afrah Alharbi (afalharbi@ksu.edu.sa)
Buthaynah Ali (buthinah.ali@gmail.com)
Aisha Ali (amali@kfmc.med.sa)
Gawaher Al-Saeed (ndaghri@ksu.edu.sa)
Nasser Al-Daghri (aldaghri2011@gmail.com)

Version: 1 Date: 27 Jul 2016

Author’s response to reviews:

Dear Dr. Heazell,

I would like to resubmit our paper entitled "Lower vitamin D levels in Saudi pregnant women are associated with higher risk of developing GDM", by Alajlan and colleagues, after addressing
the editor’s comments. We have revised the methods section accordingly and complied with the rest of the instructions.

We hope the manuscript is now ready for peer-review.

Best regards,

Nasser